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	CRUISE: EX10-04 Leg II
Bitung to Bitung
	PERSONNEL ON BOARD:

NOAA OER:

1. Mashkoor Malik (ERT Inc)

2. Dave Lovalvo (Eastern Oceanics)

3. Webb Pinner (2020 LLC)

4. Jeremy Potter (OER)

Others:

5. Doug Jongeward (PMEL)

6. John Sherrin (UVic)

7. Xerandy (BPPT)

8. Major Muddan (TNI)

9. Dave Wright (UCAR/JOSS)

10. Joel DeMello (UCAR/JOSS)

11. Tom Kok (UCAR/JOSS)

12. Tom Pierce (UCAR/JOSS)

13. Karl McLetchie (UCAR/JOSS)

14. Brian Bingham (UCAR/JOSS)
15. Randy Prickett (UCAR/JOSS)

16. Eric Prechtl (UCAR/JOSS)

17. Roland Brian (UCAR/JOSS)

18. Joe Biscotti (UCAR/JOSS)

	DATE/TIME FILED:  07/4/2010 0843 UTC

FILED BY: Jeremy Potter
	

	VESSEL:  NOAA Ship Okeanos Explorer (EX)


	

	GEOGRAPHIC AREA:  Operating north of North Sulawesi, Indonesia 
	


=============================================================================

SUMMARY:    The Okeanos Explorer Program completed ROV dives #4 and #5 on July 2nd and 3rd.  Everything continues to go quite well from both the operations and science perspectives.
Onboard operations included multibeam data acquisition, 4 CTD casts, and 2 ROV dives. Jakarta and Seattle ECCs were fully staffed during the ROV dives. 

ROV video and still imagery from dives 001 and 002 were successfully transferred to shore. 

NOTE: As of today, local ship time is 8 hrs ahead of UTC.  Jakarta is 7 hrs ahead of UTC. East Coast is 4 hrs behind UTC. West Coast is 7 hrs behind UTC.
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ROV:

July 2: The launch target for EX-1004-Leg II_ROV004  was 4º 00.760’N 125º 16.800’E. The vehicle descended to a depth of 1600m on a small seamount that lay at the base of a larger seamount to the east. It reached the bottom approximately 40m downslope on the western flank of an east-west trending ridge. The bottom for most of the dive was light coloured, sandy sediment. A few rocky outcrops appeared in areas of steeper terrain. There was no evidence of hydrothermal activity. Biology appeared dominated by sparse assemblages of solitary corals, shrimp, galatheid crabs, and sponges. Small crustaceans or barnacles were visible on several corals and sponges.

Image: Frame grab of an unidentified squid from ROV Dive 003 on July 1.
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July 3: The launch target for EX-1004-Leg II_ROV005  was 3º 46.73’’N 125º 22.18’E. The target is nearby the site termed ‘Naung’ in the 2004 McConachy paper. Multibeam bathymetry revealed a shallow seamount with a central peak and several surrounding lower peaks. CTD casts the previous night indicated hydrothermal activity. The ROV landed downslope of the central peak but was unable to make forward progress in the desired direction due to strong NE to SW current. ROV Operations Coordinator Dave Lovalvo informed the science party of the need to change tactics. He directed the ROV navigator to have the ship ‘tow’ the ROV to a new location up-current to begin the dive. Currents were greatly reduced in the new area and the ROV was able to make headway. The soft sedimented bottom was covered with many ripples and there were often large scour marks around the many rocky outcrops we encountered. The biota of Naung was similar to what we saw in our previous dive, however, this seamount contained a much greater abundance and diversity of organisms. Though the ROV found no evidence of hydrothermal activity, it did capture stunning high-resolution imagery of the various corals, sponges, shrimp, sea cucumbers, and other animals. 

Image: EX-produced 20m grid cell bathymetric map of Naung site.
There have been no additional developments associated with the USBL transceiver. This is an issue that must be addressed during the winter import. 
SURVEY/CTD:
Immediate after ROV recovery on July 2nd, survey completed four CTD stations around the Naung site: 
1. 125 º 22.13’E, 3 º 46.41’N, 600m,
2. 125 º 22.18’E, 3 º 46.33’N, 624m depth
3. 125 º 22.27’E, 3 º 46.26’N, 533m depth
4. 125 º 22.44’E, 3 º 46.18’N, 802m depth

Multibeam mapping operations started by 0000 on July 3rd  and continued until ROV launch at 0815. The survey continued through the evening of July 3rd after ROV recovery. 
SHIP NETWORK

Low-speed network – Performance has improved compare to the beginning of EX-10-04 Leg1 but continues to be poorer than EX-10-02.  The cause of the poor performance is perhaps related to an unknown source of inbound traffic.  OMAO shore-side support personnel are aware of the issue and will continue working to identify the cause and solution on Tuesday local time. 
High-speed Network – On June 20, 2010 bandwidth to the high-speed link was increased from 10Mbps to 20 Mbps per the NOAA contract with MTN. OER pays $6K per day for the service to increase the size and quality of the video feeds to the ECCs.  When the bandwidth was expanded, the ECCs experienced a serious degradation in video quality and internet access through the visitor’s network was completely lost.  Once the total size of the all three feeds was reduced to less than 10Mbps the video quality reverted back to previous levels and internet access for the visitors returned.  The ETs have confirmed with MTN that the link to the Okeanos Explorer is 20Mbps and the OMAO network engineer has confirmed the system should support outbound traffic up to 20 Mbps.  The problem appears to be linked to the ship’s TurboIP box.  The TurboIP box is responsible for accelerating TCP-based traffic (i.e web, ftp) over the satellite link. With the TurboIP box removed the ship is able to transmit video up to 20Mbps but ftp transfers fail.  MTN is contacting the manufacturer for additional assistance.  Though OMAO is responsible for solving the problem, OER is responsible for paying all MTN contract costs. To date, OER has spent $78K for services that it is unable to use because of the unknown problem. 
OMAO and OER discussions on network improvements continue. The Chief ET ceased trying to keep hours that overlap with Silver Spring personnel. Communication and testing across each group continues to be difficult due to varying time zones. OMAO and OER are arranging for a contract network engineer to examine and trouble-shoot the network from shore. 

VSAT

MTN and OMAO are looking into the >10MBps bandwidth issues described above.

Dynamic Positioning

DP system continues to work quite well during ROV operations. There are no issues to report.
OTHER:

Chief ET Conway temporarily fixed the interface to the satellite phone. A replacement part is on order to permanently remedy the problem.
============================================================================================================================================================
PLAN OF THE DAY

Sunday, July 04, 2010

OPERATIONS: Exploration Indonesia

	0000
	Underway as before
Conducting mapping operations

	0745
	Safety Brief (Bridge)

	1500
	Ops Brief (Forward Lounge)

	1830
	Fourth of July Sparklers Party (Bow)


· Continue multibeam operations
· 0030, 0630, 1230, 1830 conduct XBT
· Pump out CHT tank
On Deck for Monday:

· Conduct ROV operations

· Possibly conduct CTD ops
· Conduct multibeam and XBT operations overnight
==============================================================================
=========================END OF SITUATION REPORT==============================
==============================================================================

